Author's response to reviews

Title: Leptin as a critical regulator of hepatocellular carcinoma development through modulation of human telomerase reverse transcriptase

Authors:

Nikolaos Stefanou (ni_stef@hotmail.com)
Vassilis Papanikolaou (papanikolaou.vassilis@gmail.com)
Yoichi Furukawa (furukawa@ims.u-tokyo.ac.jp)
Yusuke Nakamura (yusuke@ims.u-tokyo.ac.jp)
Aspasia Tsezou (atsezou@med.uth.gr)

Version: 7 Date: 18 August 2010

Author's response to reviews:

To Dr. Robin Cassady-Cain

on behalf of

Rachel Neilan, MSc
Scientific Editor
August 18, 2010

Dear Sirs

Thank you for your emails informing us about the acceptance of our manuscript MS 1636314955321586 entitled: “Leptin as a critical regulator of hepatocellular carcinoma development through modulation of human telomerase reverse transcriptase”.

We wish to submit a revised version of our manuscript, where two typing errors have been corrected: page 8, paragraph 2, title, second line, we replaced "mediated" by "mediated" and page 9, paragraph 2, line 3, where we omitted an extra space between "of" and "IL-6".

These two changes are not highlighted in the manuscript.

Thank you for your co-operation.

Sincerely,

Aspasia Tsezou
Assoc.Prof. Med. Genetics
University of Thessaly, Medical School